Author’s response to reviews

Title: Vitamin D Receptor Gene Associations with Pulmonary Tuberculosis in a Tibetan Chinese population

Authors:

Qunying Hu (huqunying15@163.com)
Zhengshuai Chen (zshuaichen@163.com)
Guinian Liang (liangguinian11@163.com)
Fangping Mo (mofangping423@163.com)
Hengxun Zhang (hengxunzhang@126.com)
Shilin Xu (xushilin116@163.com)
Yuhe Wang (wangyuhe10@sina.com)
Longli Kang (klongli@163.com)
Tianbo Jin (jtianbo@126.com)

Version: 3 Date: 06 Jul 2016

Author’s response to reviews:

Dear Editor,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate you very much for your positive and constructive comments and suggestions on our manuscript entitled “Vitamin D Receptor Gene: Association with Pulmonary Tuberculosis in a Tibetan Chinese population”. (Manuscript ID: INFD-D-15-00216R2).

We have carefully evaluated the critical your comments and thoughtful suggestions, responded to these suggestions point-by-point, and revised the manuscript accordingly. All authors have agreed to authorship and order of authorship for this manuscript and have the appropriate permissions and rights to the reported data. Revised portion are marked using the track changes
mode in MS Word. The main corrections in the paper and the responds to the editor’s comments are as follows:

Editorial Revisions

1、Please check your manuscript carefully as there were a few language errors remaining (mainly in the last set of revisions) and at least one place where AA was used instead of A/A

Response: Thanks for the reviewer’s kind suggestion. We have checked our manuscript carefully and corrected several language errors in our revised manuscript.

2、Please ensure that your manuscript contains a Declarations section before the references, with ALL of the following subsections (at present the ethics sub-section is missing):

   * List of abbreviations
   * Ethics approval and consent to participate
   * Consent for publication – as you are not publishing identifying patient data, you can write "Not applicable" in this section
   * Availability of data and materials
   * Competing interests
   * Funding
   * Authors' contributions
   * Acknowledgements.

Response: Thank you very much for your reminder. ALL of the above subsections were contained in our manuscript.

We tried our best to improve the manuscript and made some changes in the manuscript. We appreciate for editor’s warm work earnestly, and hope that the correction will meet with approval. Once again, thank you very much for your suggestions.
Best regards!

Yours sincerely,

Tianbo Jin

E-mail: jitianbo@126.com